CSE548 Multiprocessors 1

Final

Include the memory system in performance Cache configuration tradeoffs Hardware & compiler techniques to hide memory latency, reduce memory latency, eliminate memory ops Hardware techniques to increase memory bandwidth Memory management with

• multiple page sizes • forward-mapped, sparse page tables Disk & bus basics Network issues & their rationale The religious war: SM vs. MP Cache coherency on bus-based & distributed MIMDs Synchronization Weak consistency models

Remember that all our machines have memory subsystems, they synchronize & have coherent caches